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| HC schedule

o From Chamonix workshop:

 Keep energy at 13 TeV for Run2 ('17-'18), go to 14 TeV for Run3

« Goal of 45 fb™! integrated luminosity with stable beams ~50% of the
time (cf. 41 fb! delivered in 2016)

 No heavy-ion run in 2017

Start powering
Controls

tests phase 1
Jan Meyenon Feb Mar

June 12: Start of stable

beams collisions
June




CMS status
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Today, find out more about the complex operation
taking place in the CMS cavern ahead of the LHC
restart this spring: the replacement of the Compact
e Barrel-pixel
(BPIX) installed
last week

Muon.SoIcno»d - CMS pixel detector, the heart of the
e Forward-pixel
(FPIX) being
Installed this
week

Forward hadron calorimeter (HF)
* |nstalled reworked PMT boxes (for dual readout)
« New readout electronics installed
Endcap hadron calorimeter (HE)
1 readout box installed and working
 Decision to be made whether to leave it in
3

4 In afew minutes, join our “heart surgeon” scientists
live for your chance 1o ask them questions.
... Meer weergeven




CMS commissioning

e Mid-Week Global Runs with Cosmics

 MWGR#1 = 08-10 February Average pulse shape in HE
» Re-established global operation of of 7 cosmiIC muon events
CMS recorded during MWGR#1

(1 photo-electron = 40 fC)

« Confirmed trigger timing

 Successfully included the new HE S
readout box (only 1 day after & 10 1 .
installation!) S ool ‘ .

« MWGR#2 = 01-03 March w0l -+ _
« MWGR#3 = 15-17 March ool :
« CRUZET (Cosmic RUnning at ZEro Tesla) wol- E
and CRAFT (Cosmic Running At Four Tesla) | + E
o 271 of March through April : L ——
e Goal: >95% efficiency with all detectors R R B A A

TS (25 ns)

* Include pixels by mid-April



Computing

 CMS production workflows in
preparation for Moriond
conference mostly completed.
Plenty of analysis jobs running

LPC usage

e Some unexpected T1 down time
(dCache down, PhEDEx down)

e Also some EOS crashes due to
\ Total Slots 4752 6792 6729 6688
increased usage by LPC users  ClaimedSiots 575 6830 5200 6197
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summary

« Extended Year End Technical Stop (EYETS) still ongoing
 Upgraded HF PMTs and readout electronics installed

 Upgrade of 1 HE readout box successful

 New pixel detector being installed
« Commissioning with cosmics is ongoing
e Preparations for Phasell upgrades ongoing

« Conceptual design proposal for MIP Timing Detector being
reviewed. CMS decision to include in upgrade scope to be
made early April.

e Analyses preparing for strong showing at Moriond conference



